Author’s response to reviews

Title: Concomitant glenohumeral pathologies in high-grade acromioclavicular separation (type III - V)

Authors:

Jochen Markel (jochen.markel@googlemail.com)
Tim Schwarting (docschwarting@me.com)
Dominik Malcherczyk (malcherc@med.uni-marburg.de)
Christian Dominik Peterlein (Christian-Dominik.Peterlein@uk-gm.de)
Steffen Ruchholtz (ruchholt@med.uni-marburg.de)
Bilal Farouk El-Zayat (elzayat@med.uni-marburg.de)

Version: 4 Date: 31 Oct 2017

Author’s response to reviews:

Dear Sirs,

Thank you very much for accepting our publication.

We performed all requested changes. Please find a point-to-point answer below:

Editor Comments:

1. Please include the Abstract section as the first page of your manuscript after the title page.
   Done

2. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.
   Done
3. Please move the Declarations to after the abbreviations section and just before the References.
Done

4. Please include a statement on ethics approval and consent to participate in the “Ethics approval and consent to participate” section of the Declarations including the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate....
Done

5. We would also like to ask for you to provide more justification for the contributions of ‘CDP’ and ‘SR’, as currently they do not automatically qualify for authorship. Contribution to helping draft the manuscript, alone, does not usually justify authorship.

The declaration of contributions was thoroughly corrected to be more precise.

6. We note that you have not included an acknowledgements section in the Declarations. If you have no acknowledgements please put ‘Not Applicable’ in this section.
Done

7. Please remove the figure legends from the figure files and instead provide the figure legends under a separate heading of 'Figure Legends' after the References.
Done

8. Please change the ‘Introduction’ heading to ‘Background’ and the ‘Materials and methods’ heading to ‘Methods’.
Done
9. Please remove the title page from the supplementary files.

Done

We looking forward for quick publication and like to

Thank you very much and best regards

Bilal El-zayat